Author's response to reviews

Title: When worlds collide

Authors:

    Clive E Adams (clive.adams@nottingham.ac.uk)
    Mauricio S de Lima (LIMA_MAURICIO_SILVA_DE@LILLY.COM)

Version: 3 Date: 21 October 2009

Author's response to reviews:

Dear Editor

RE: MS: 1730735368294736 - When worlds collide

I am so sorry to have been so tardy in responding to your corrections and comments. I am also sorry for submitting the manuscript with misspellings. I have corrected everything and responded to every comment. The peer review was helpful – thank you.

I was not quite sure what to put in response to your suggestion that we put a couple of lines in about why we wrote this. I have done so right at the start and hope that is OK. The article we put together because of our friendship – partly from conversations we had had across the years. I work in a culture where collaboration with industry is unacceptable because of the accompanying biases – when that is only true because of rather bad behaviour on the parts of both parties – but this is also a way by which other biases go un-tackled.

Thank you for your help with this paper.

Yours sincerely

Clive E Adams

Professor Clive E Adams
Co-ordinating Editor Cochrane Schizophrenia Group
Chair of Mental Health Services Research,
Division of Psychiatry, Sir Colin Campbell Building,
Jubilee Campus,
University of Nottingham Innovation Park,
Triumph Road, NG7 2TU, UK
E-mail: mailto:clive.adams@nottingham.ac.uk
Tel: +44 (0)115 82 31 274
Skype: ceadams2982